Author's response to reviews

Title: The roles of tumor necrosis factor-alpha in colon tight junction protein expression and intestinal mucosa structure in a mouse model of acute liver failure

Authors:

Li Hong Song (hlsong26@yahoo.com.cn)
Sa Lv (lysa@sina.com)
Pei Liu (syliupei2003@yahoo.com.cn)

Version: 5 Date: 17 May 2009

Author's response to reviews:

1. We write our paper to use the Times New Roman style of written English.

2. We revise our manuscript to add these two articles (PMID: 15968724 and 17663824) and discuss them.

3. We provide a statement on animal ethics, sections with competing Interests and Authors' contributions.

4. We revise our manuscript by a native English speaker and make it easier to read.